Author’s response to reviews

Title: Experience of maintaining tooth brushing for children born with a cleft lip and/or palate

Authors:
Yin-Ling Lin (yin-ling.lin@manchester.ac.uk)
Karen Davies (karen.davies-3@manchester.ac.uk)
Peter Callery (peter.callery@manchester.ac.uk)

Version: 1 Date: 10 Aug 2016

Author’s response to reviews:
Dear Editor,

Thank you for your comment. A statement about ethics approval and consent to participate had already been included in declarations, please see P.10 of the manuscript. We have also included the information about the ethical approval in the main text in the revised manuscript (P.4).

Yours Sincerely,
Yin-Ling